Author’s response to reviews

Title: Relapsing Optic Neuritis and Meningoencephalitis in a Child: Case report of Delayed Diagnosis of MOG-IgG Syndrome

Authors:

Xiaonan Zhong (zhong_zxn@163.com)
Yanyu Chang (changyyu@163.com)
Sha Tan (tsts926jay@aliyun.com)
Jingqi Wang (amanda0218@yeah.net)
Xiaobo Sun (sister_555@163.com)
Aimin Wu (wuaim@126.com)
Lisheng Peng (lishengpeng79@163.com)
Alexander Lau (alexlau@cuhk.edu.hk)
Allan Kermode (allan.kermode@uwa.edu.au)
Wei Qiu (qiuwei120@vip.163.com)

Version: 1 Date: 28 Apr 2019

Author’s response to reviews:

We thank the reviewers for their encouraging comment of this article and very careful review. The questions and advice were very helpful for improving the quality of our article. We addressed the points accordingly.

1. Which virus types were scanned in virus antibody screening on blood sample? (In page 3 line 48)

Microorganism (Toxoplasma, Rubella Virus, Cytomegalovirus, Herpes Simplex Virus, Epstein-Barr Virus) antibodies were screened on blood sample.

2. Had viral PCR (Polymerase chain reaction) screening in cerebrospinal fluid been performed during meningoencephalitis attacks? (In page 4)
Microbial second generation sequencing was conducted at the last meningoencephalitis attacks when she 14 years old.